Author's response to reviews

Title: Autonomy among physically frail older people in nursing home settings: a study protocol for an intervention study.

Authors:

Mette MA Andresen (mea@ucsj.dk)
Lis LP Puggaard (lipu@cowi.dk)

Version: 2 Date: 14 October 2008

Author's response to reviews:

At the first page in the main text the first word "Abstract" must be deleted.